Author's response to reviews

Title: Immunohistochemical profiling of benign, low malignant potential and low grade serous epithelial ovarian tumors

Authors:

Veronique Ouellet (ouelletv@yahoo.ca)
Tak Hay Ling (takhay7777@yahoo.ca)
Karine Normandin (ka_normandin@hotmail.com)
Jason Madore (jason.madore@gmail.com)
Christian Lussier (clu2001@videotron.ca)
Veronique Barres (veronique.barres@usherbrooke.ca)
Dimcho Bachvarov (dimtcho.bachvarov@crhdq.ulaval.ca)
Claudine Rancourt (claudine.rancourt@usherbrooke.ca)
Patricia N Tonin (patricia.tonin@mcgill.ca)
Diane M Provencher (diane.provencher.chum@ssss.gouv.qc.ca)
Anne-Marie Mes-Masson (anne-marie.mes-masson@umontreal.ca)

Version: 4 Date: 19 November 2008

Author's response to reviews: see over
Dr Sabina Alam  
Assistant Editor  
BMC-series journals

Please find included the modifications we have brought to the manuscript in order to include the formatting request.

**Change all headings to sentence case with only the first letter of the first word capitalised.**

This modification was made.

**Abbreviations - please format the abbreviations as a paragraph rather than as a list.**

This modification was made.

**Reference 33-36 not cited in correct order.**

This modification was made.

Corresponding author
Anne-Marie Mes-Masson, Ph.D.
CR-CHUM/ICM
1560, rue Sherbrooke est
Montreal, Quebec, Canada H2L 4M1
Telephone: 514-890-8000 ext 25496
Fax: 514-412-7703
E-mail: anne-marie.mes-masson@umontreal.ca